Author's response to reviews

Title: Cost-minimization analysis of the direct costs of TPE and IVIg in the treatment of Guillain-Barre syndrome

Authors:

Jeffrey L Winters (winters.jeffrey@mayo.edu)
David Brown (DBrown1@hne.com)
Elisabeth Hazard (hodiehazard@yahoo.com)
Ashok Chainani (Ashok.Chainani@caridianbct.com)
Chester Andrzejewski Jr (chester.andrzejewski@baystatehealth.org)

Version: 5 Date: 11 May 2011

Author's response to reviews:

To whom it may concern,

We have corrected the issues described in the email dated 5/11/11. This included cutting and pasting the abstract from the current manuscript into the submission form.

Please note that these issues referenced in the 5/11/11 email were not included in the email dated 4/28/11. We have now removed ALL lines from the tables, including the borders. In addition to the requested changes, some typographical corrections, including one misspelling and two missing commas, were made.

Sincerely,

Jeffrey L. Winters, M.D.